

# **Original citation:**

Courtenay, Orin, Carson, Connor, Calvo-Bado, Leo A., Garcez, Lourdes M. and Quinnell, Rupert J.. (2014) Heterogeneities in Leishmania infantum Infection: Using Skin Parasite Burdens to Identify Highly Infectious Dogs. PLoS Neglected Tropical Diseases, Volume 8 (Number 1). Article number e2583. ISSN 1935-2735

# Permanent WRAP url:

http://wrap.warwick.ac.uk/58744

# Copyright and reuse:

The Warwick Research Archive Portal (WRAP) makes this work of researchers of the University of Warwick available open access under the following conditions.

This article is made available under the Creative Commons Attribution 3.0 (CC BY 3.0) license and may be reused according to the conditions of the license. For more details see: <u>http://creativecommons.org/licenses/by/3.0/</u>

# A note on versions:

The version presented in WRAP is the published version, or, version of record, and may be cited as it appears here.

For more information, please contact the WRAP Team at: publications@warwick.ac.uk



highlight your research

http://wrap.warwick.ac.uk

# Heterogeneities in *Leishmania infantum* Infection: Using Skin Parasite Burdens to Identify Highly Infectious Dogs

# Orin Courtenay<sup>1,9</sup>\*, Connor Carson<sup>1,9</sup>, Leo Calvo-Bado<sup>1</sup>, Lourdes M. Garcez<sup>2,3</sup>, Rupert J. Quinnell<sup>4</sup>

1 School of Life Sciences, and Warwick Infectious Disease Epidemiology Research (WIDER), University of Warwick, Coventry, United Kingdom, 2 Laboratório de Epidemiologia e Imunologia aplicada às Leishmanioses, Seção de Parasitologia, Instituto Evandro Chagas, Belém, Pará, Brazil, 3 Centro do Ciências Biológicas e da Saúde, Universidade do Estado do Pará, Belém, Pará, Brazil, 4 School of Biology, University of Leeds, Leeds, United Kingdom

# Abstract

**Background:** The relationships between heterogeneities in host infection and infectiousness (transmission to arthropod vectors) can provide important insights for disease management. Here, we quantify heterogeneities in *Leishmania infantum* parasite numbers in reservoir and non-reservoir host populations, and relate this to their infectiousness during natural infection. Tissue parasite number was evaluated as a potential surrogate marker of host transmission potential.

*Methods:* Parasite numbers were measured by qPCR in bone marrow and ear skin biopsies of 82 dogs and 34 crab-eating foxes collected during a longitudinal study in Amazon Brazil, for which previous data was available on infectiousness (by xenodiagnosis) and severity of infection.

*Results:* Parasite numbers were highly aggregated both between samples and between individuals. In dogs, total parasite abundance and relative numbers in ear skin compared to bone marrow increased with the duration and severity of infection. Infectiousness to the sandfly vector was associated with high parasite numbers; parasite number in skin was the best predictor of being infectious. Crab-eating foxes, which typically present asymptomatic infection and are non-infectious, had parasite numbers comparable to those of non-infectious dogs.

*Conclusions:* Skin parasite number provides an indirect marker of infectiousness, and could allow targeted control particularly of highly infectious dogs.

Citation: Courtenay O, Carson C, Calvo-Bado L, Garcez LM, Quinnell RJ (2013) Heterogeneities in *Leishmania infantum* Infection: Using Skin Parasite Burdens to Identify Highly Infectious Dogs. PLoS Negl Trop Dis 8(1): e2583. doi:10.1371/journal.pntd.0002583

Editor: Marleen Boelaert, Institute of Tropical Medicine, Belgium

Received August 6, 2013; Accepted October 29, 2013; Published January 9, 2014

**Copyright:** © 2014 Courtenay et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** The study received financial support from the Wellcome Trust (grant numbers 036365 and 069169, www.wellcome.ac.uk) to OC and RJQ. CC was supported by a BBSRC Case Studentship (www.bbsrc.ac.uk). The funders had no role in study design, data collection and analysis, decision to publish or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: o.courtenay@warwick.ac.uk

9 These authors contributed equally to this work.

# Introduction

Studies of microparasites usually consider hosts as homogeneous infection units (infected or uninfected), despite knowledge that infections progress through states of clinical severity, that clinical severity is often associated with the number of infecting microorganisms (load), and that individual transmission potential may be related to infection load. The significance of "super-spreaders" responsible for spreading infection to a disproportion-ate number of secondary cases has long been recognised [1,2], however the relationships between parasite load and transmission are rarely measured; even in well-studied macroparasites (e.g. helminths) infectiousness is assumed to correspond to worm burden and egg count [3–6].

Variations in individual infection loads tend to be characterised by right-skewed (over-dispersed or aggregated) frequency distributions. Over-dispersion translates into diminishing proportions of the host population harbouring disproportionately higher infection loads. Where transmission potential is directly related to infection load, over-dispersed distributions may be interpreted as a small fraction of the population being responsible for most transmission, giving rise to the "20/80 rule" (whereby 20% of cases cause 80% of transmission), proposed for a number of parasitic agents (e.g. [7–10]). Heterogeneity in transmission can increase the basic case reproduction number  $R_0$  of a pathogen compared to that under assumptions of homogeneous mixing or density-dependent contact networks [9,11], and affect the effort required, and choice of strategy (mass or targeted), to interrupt transmission [7–9,12]. Molecular techniques, such as real-time quantitative PCR (qPCR), have been used recently to differentiate between infected individuals and to help understand the spread and treatment of emerging infectious diseases e.g. [2,9,13–15], nevertheless few empirical studies relate individual infection loads to transmission.

Zoonotic visceral leishmaniasis (ZVL) is a fatal disease of humans and canids caused by the protozoan parasite *Leishmania infantum*, and transmitted between hosts by Phlebotomine sandflies. The domestic dog is the only proven reservoir [16], though severity of infection and infectiousness varies greatly between individuals; in humans and wild mammals the majority of

## **Author Summary**

Zoonotic visceral leishmaniasis is a sandfly-borne disease of humans and dogs caused by the intracellular parasite Leishmania infantum. Dogs are the proven reservoir. The disease is usually fatal unless treated, and is of global health significance. Diagnosis of canine infections relies on serum antibody-based tests that measure infection. In some endemic regions, a test-and-slaughter policy of seropositive dogs forms part of the national control policy to reduce human infection. However, this strategy is not considered effective. Since not all infected dogs are infectious to sandfly vectors, one option is to target control at infectious dogs, as only these dogs maintain transmission. We quantify Leishmania numbers in individual host tissues from time of infection using molecular methods. Comparing these results with their infectiousness to sandflies, we also evaluate the performance of molecular and immunological assays to identify infectious animals. Parasite numbers varied substantially between individuals, increasing with duration and severity of disease. Infectiousness to the sandfly vector was associated with high parasite numbers, and parasite loads in the skin was the best predictor of being infectious. The results suggest that molecular quantitation is useful in identifying individuals and populations responsible for maintaining transmission, with potential application in operational control programmes.

infections are asymptomatic and non-infectious [16]. Control of ZVL focuses on the detection and elimination of infected dogs (particularly in South America), indoor residual spraying of insecticide, and human case treatment [17]. Positivity to serum anti-Leishmania antibodies is the principal criterion for mandatory slaughter of dogs [17]. Analyses indicate that this policy has little impact on reducing ZVL incidence, though robust data are lacking [16], and there have been calls to re-evaluate the ZVL control program in Brazil [16,18-21]. Contributing factors to the lack of effectiveness include delays between testing and slaughter, low test sensitivity [22], and significant dog-owner non-compliance [21]. An alternative strategy could be to target infectious rather than infected dogs, providing infectious hosts can be identified. Direct measurement of infectiousness by xenodiagnosis requires blood-feeding of colony-reared sandflies on hosts followed by screening for parasite infections in the vector. Rearing large quantities of vectors for community surveillance however is not practical. Tissue parasite loads have the potential to provide a reliable indirect marker of infectiousness [23-30], though no studies have tested these relationships through the time course of infection.

Here we measure *L. infantum* loads in cohorts of naturally infected domestic dogs *Canis familiaris* and crab-eating foxes *Cerdocyon thous* in Amazon Brazil. This study is unique in being able to relate host tissue parasite loads to serial xenodiagnosis from time of natural infection. The aims were (i) to characterize the heterogeneities in *L. infantum* loads between sampled tissues and between individual hosts with different severity of infection, (ii) to investigate whether tissue parasite loads can predict infectiousness to the sandfly vector; (iii) to compare parasite loads between dogs and crab-eating foxes, and (iv) to evaluate the performance of qPCR and ELISA diagnostic assays to identify infectious animals in mixed populations.

## **Materials and Methods**

#### Ethics statement

Canine samples were collected with informed consent from dog owners. Sampling was performed in accordance with UK Home Office guidelines.

### Study site and study design

Dog samples were available from  $-80^{\circ}$ C archived material generated in a cohort study of naturally exposed dogs between April 1993 and July 1995 in the municipality of Salvaterra, Marajó Island, Pará State, Brazil, in which bone marrows aspirated from the iliac crest and 3 mm skin biopsy punches of the ear pinnae outer edge were sampled repeatedly at approximately 2 month intervals for up to 27 months post initial exposure [31]. Ear skin was the preferred skin sample since it is reported to be more infectious to sandflies than abdomen skin [23,30]. Both skin and bone marrow are reported to be more sensitive than blood for parasitological and molecular detection of L. infantum, and higher qPCR counts are recorded in bone marrow than in blood [32-34]. For the present study, 265 bone marrow samples were available from 82 infected dogs (1-10 samples per dog), and 185 ear skin biopsy samples were available from 64 infected dogs (1-6 samples per dog), of which 173 samples from 63 dogs had paired bone marrow samples. Fox samples were collected during a concurrent longitudinal study of sympatric marked-recaptured free-ranging foxes [35]. Here, 67 bone marrow samples from 34 infected foxes, and 51 ear biopsy samples from 30 infected foxes, were available; all ear biopsy samples had paired bone marrow samples. Dog samples were collected with informed consent from dog owners.

## Assays

Dog and fox samples were assayed at all, or at the majority, of time-points, for (i) anti-Leishmania IgG by ELISA using crude leishmanial antigen (CLA), with antibody concentrations expressed as arbitrary units/mL relative to a positive control serum [31] (n = 277 samples); (ii) PCR on bone marrow biopsies using primers specific for kinetoplast DNA (kDNA) and ribosomal RNA [36] (n = 277 samples); (iii) rK39 Kalazar Detect Rapid Diagnostic Test (RDT), Inbios International Inc., WA., USA [37], (iv) qPCR primers for kDNA (described below), and (v) clinical score, defined as the sum of the score of six typical clinical signs (alopecia, dermatitis, chancres, conjunctivitis, onychogryphosis, and lymphadenopathy), each scored on a semi-quantitative scale from 0 (absent) to 3 (intense) [36] (n = 266 samples). Animals were assessed for infectiousness to the sandfly vector by xenodiagnosis, using uninfected colony-reared Lutzomyia longipalpis, and following dissection 4–5 days post full engorgement [22,35]. Here, matching xenodiagnosis data were available for 103 dog bone marrow samples (36 infected dogs, 3,751 fed flies dissected), 58 dogs ear samples (26 infected dogs, 1,702 flies), 39 fox bone marrow samples (22 infected individuals, 1,309 flies), and 30 fox ear samples (18 foxes, 1,187 flies).

## Quantitative PCR (qPCR)

DNA was extracted from 100  $\mu$ L aliquots of bone marrow, using phenol-chloroform [38]. DNA from 3 mm ear skin punch biopsies (average: 0.029 grams, range: 0.0144–0.0837) was extracted using a commercial kit (DNeasy: Qiagen, UK). qPCR was performed using primers specific for a conserved region of *Leishmania* kDNA [27]. Quantification of *Leishmania* DNA was performed by comparison of C<sub>t</sub> values with those from a standard curve constructed from 10-fold dilutions of *L. infantum* DNA extracted from cultured parasites, from 1×10<sup>5</sup> to 0.001 parasite

equivalents/mL (strain MHOM/MA/67/ITMAP-263). Samples were tested in duplicate and standards in triplicate on every plate. The occasional duplicates giving one positive and one negative result were re-tested: none remained unresolved after re-testing. A non-template control (NTC) was run in triplicate on every plate. A plate of negative controls including DNA extracted from blood samples of 30 UK dogs with no history of foreign travel, and 40 endemic control dogs from São Paulo, Brazil was run every 5 plates. A standardised Ct threshold value of 0.01 was selected as cut-off value to define infection based on the NTC signal. The endogenous control was a eukaryotic 18S rRNA gene as a reference of total canine DNA quantified in a separate qPCR reaction to the Leishmania assay using pre-developed TaqMan Assay reagents (Applied Biosystems, UK) following the manufacturer's recommendations. Parasite loads were normalized (d) between animals to the eukaryotic 18S rRNA gene per reaction, where d = absolute Leishmania kDNA equivalents/(copy number of 18S rRNA gene/2)/ng tissue DNA extracted measured spectrophotometrically. Normalized  $log_{10}$  parasite numbers and absolute log<sub>10</sub> parasite numbers per ml (bone marrow) or per gram (ear skin) were strongly correlated ( $r^2 = 0.93$  and  $r^2 = 0.98$  respectively). Consequently, for ease of interpretation, we report the per unit absolute  $\log_{10}$  parasite numbers.

#### Definition of infection and infectiousness

The date of patent infection for dogs and foxes was estimated as the first date at which animals were positive by any serological or parasitological assay; all samples thereafter were considered as infected based on previous analyses demonstrating a very low incidence of serological reversal [31,35,36]. At each bimonthly examination, dogs were classified according to their total clinical score as asymptomatic (scores 0-2), oligosymptomatic (3-6) and symptomatic (>6). Dogs with >8 months post infection follow-up and all bimonthly clinical scores <3 were considered long-term asymptomatic. Infectiousness was assessed as either positive ( $\geq 1$ sandfly infected) or negative, or as the proportion of sandflies infected at any single time point (point xenodiagnosis). Dogs were also classified previously [22,35] as "highly infectious" (>20% of total flies infected), "mildly infectious" (>0% and <20% flies infected), and "non-infectious" (no flies infected) by serial xenodiagnoses (n = 6,002) flies dissected from 173 independent trials): the highly infectious group were shown to be responsible for >80% of all transmission events [22]. All foxes were noninfectious (n = 1,469 flies from 44 trials) [35].

#### Data analysis

Parasite aggregation was characterised by the dispersion coefficient k of the fitted negative binomial distribution. Negative binomial models were used to test for differences in parasite loads between groups. Analysis of parasite loads against independent variables were conducted using negative binomial mixed models, with animal identity included as the random effect. The relationship between infectiousness and markers of infection was analysed by logistic regression.

Receiver Operating Curves (ROC) were used to identify parasite load (qPCR) and anti-*Leishmania* antibody (ELISA) threshold values that maximised test sensitivity and specificity to differentiate currently infectiousness and non-infectious dogs. Areas under the ROCs were similar: 0.937 (ear biopsies, n = 58), 0.837 (bone marrows, n = 103) and 0.846 (ELISA, n = 173) ( $\chi^2$  = 72.0, df = 2, P = 0.699, n = 52), providing test threshold values of 4.64 log<sub>10</sub> parasites/gram (ear biopsies), 3.51 log<sub>10</sub> parasites/mL (bone marrows), and 4.59 log<sub>10</sub> antibody units/mL, respectively. These values were then used to evaluate the performance of threshold-based qPCR and ELISA assays to detect dogs classified by longitudinal infectious status in the mixed population. The average times of detection by the threshold-based assays relative to infection were calculated using Kaplan-Meier survival analysis. Differences in Kaplan-Meier curves were compared by log rank test, and confidence limits calculated following [39]. All analyses were carried out in Stata v.11.1 (Stata Corporation, College Station, Texas, USA).

## Results

#### Leishmania loads of infected dogs

Parasite loads were quantified by qPCR in 265 post-infection bone marrow samples from 82 dogs, and 185 post-infection ear skin biopsies from 64 dogs (Table 1). The median parasite loads were 142 parasites/mL in bone marrow and 119 parasites/gram in ear skin (Table 1) but the correlation was not strong (Spearman's  $\rho = 0.56$ , P<0.001). Note that since the unit of measurement of these two samples differ, the magnitude of the parasite loads in skin and bone marrow were not directly compared. The frequency distributions of parasite loads in both tissues was highly skewed, with maximum burdens of  $2.4 \times 10^6$ parasites/mL and 1.3×10<sup>8</sup> parasites/gram in bone marrow and ear skin, respectively (Figure 1). The degree of parasite aggregation, measured by the negative binomial parameter k, was very high, with loads in ear skin (k = 0.066) showing greater aggregation than those in bone marrow (k=0.104). Comparable degree of aggregation was observed for mean parasite loads in individual dogs (Table 1). Of the total L. infantum loads recorded in bone marrows biopsies, 90% of parasites were found in 8% (21/265) of

**Table 1.** Comparison of the *L. infantum* parasite loads, and their degree of aggregation, in sympatric dog and fox populations in Amazon Brazil.

|                             | n                           | mean                | median | IQR      | range                 | <i>k</i> (95% CL)      |  |  |
|-----------------------------|-----------------------------|---------------------|--------|----------|-----------------------|------------------------|--|--|
| sample                      | samples from infected dogs  |                     |        |          |                       |                        |  |  |
| ВМ                          | 265                         | 1.0×10 <sup>5</sup> | 142    | 4–4194   | 0-2.4×10 <sup>6</sup> | 0.104<br>(0.091–0.120) |  |  |
| skin                        | 185                         | 2.7×10 <sup>6</sup> | 119    | 0–8670   | 0-1.3×10 <sup>8</sup> | 0.066<br>(0.056–0.079) |  |  |
| means                       | means for each infected dog |                     |        |          |                       |                        |  |  |
| BM                          | 82                          | 1.3×10 <sup>5</sup> | 1712   | 18–23835 | 0-2.2×10 <sup>6</sup> | 0.138<br>(0.108–0.175) |  |  |
| skin                        | 64                          | 3.9×10 <sup>6</sup> | 412    | 16–17936 | 0-1.3×10 <sup>8</sup> | 0.077<br>(0.058–0.102) |  |  |
| sample                      | s from                      | infected fo         | xes    |          |                       |                        |  |  |
| ВМ                          | 67                          | 2.9×10 <sup>4</sup> | 0      | 0–1744   | 0-1.1×10 <sup>6</sup> | 0.042<br>(0.029–0.062) |  |  |
| skin                        | 53                          | 8.4×10 <sup>4</sup> | 0      | 0–349    | 0-1.8×10 <sup>6</sup> | 0.047<br>(0.031–0.070) |  |  |
| means for each infected fox |                             |                     |        |          |                       |                        |  |  |
| BM                          | 34                          | 2.5×10 <sup>4</sup> | 12     | 0–2167   | 0-5.3×10 <sup>5</sup> | 0.057<br>(0.034–0.094) |  |  |
| skin                        | 30                          | 6.3×10 <sup>4</sup> | 41     | 0–2893   | 0-8.3×10 <sup>5</sup> | 0.076<br>(0.048–0.121) |  |  |

Statistics shown for individual samples and for means of all samples from each animal. Parasite values are expressed as number/mL (BM bone marrow) and number/gram (ear skin). IQR interquartile range; *k* negative binomial over-dispersion statistic (CL confidence limits). doi:10.1371/journal.pntd.0002583.t001



Figure 1. Frequency distributions of *L. infantum* parasite numbers in dog tissue samples measured by qPCR. Parasite numbers are expressed as numbers per mL (bone marrow, black bars) or per gram (ear skin, light bars). doi:10.1371/journal.pntd.0002583.g001

samples and 16% (13/82) of dogs; for skin biopsies, the equivalent figures were 8% (14/185) of samples and 9% (6/64) of dogs.

Parasite loads in both tissues increased on average with time since infection (Table 2; Figure 2). Ear skin loads increased at a faster average rate than bone marrow loads, reflected in the ear skin to bone marrow parasite load ratios being significantly greater in later infection (Table 2). However, the relationship between parasite load and time varied between individual dogs, showing positive to negative slopes for both tissues (Figure 3). Both bone marrow and ear skin loads were significantly higher in sick dogs, in infectious dogs and in dogs with higher anti-*Leishmania* antibody levels (Table 2). Severity of infection was also associated with greater ear skin to bone marrow parasite ratios (Table 2). However, in symptomatic dogs this ratio did not vary according to the type of symptom: dogs with skin symptoms had comparable ratios to those with only non-skin symptoms (IRR = 0.67 (95% CL 0.28–1.62),  $\chi^2 = 0.79$ , P = 0.37).

#### Leishmania loads and infectiousness to sandflies

The probability of a dog being infectious to sandflies at point xenodiagnosis was positively associated with parasite load, PCR status, IgG antibody titer, total clinical score, and time since

**Table 2.** Relationship between dog tissue *L. infantum* loads, time since infection, anti-*Leishmania* IgG, clinical status, and infectiousness.

| Variable                                      | Bone marrow   | Ear skin  | Ear skin/bone ratio <sup>1</sup>                                |
|---|---|---|---|
| Time infection (days)                         | 1.001 (1.000–1.002) N = 265 Wald $\chi^2$ = 7.95, df = 1 P = 0.0048 | 1.003 (1.002–1.004) N=185 Wald $\chi^2$ =27.3, df=1 P<0.0001    | 1.003 (1.002–1.004) N=173 Wald $\chi^2\!=\!26.6,$ df=1 P<0.0001 |
| Anti- <i>Leishmania</i> IgG<br>(log units/mL) | 1.54 (1.31–1.82) N=265 Wald $\chi^2$ =26.5, df=1 P<0.0001           | 2.55 (1.96–3.32) N = 185 Wald $\chi^2$ = 48.7, df = 1 P<0.0001  | 2.11 (1.59–2.81) N=173 Wald $\chi^2$ =26.6, df=1 P<0.0001       |
| Infectious by<br>xenodiagnosis <sup>2</sup>   | 2.20 (1.25–3.88) N = 58 Wald $\chi^2$ = 7.39, df = 1 P = 0.0066     | 4.27 (1.75–10.4) N = 58 Wald $\chi^2$ = 10.2, df = 1 P = 0.0014 | 1.69 (0.57–4.98) N=52 Wald $\chi^2$ =0.91, df=1 P=0.34          |
| Clinical status <sup>3</sup>                  |   |   |   |
| Asymptomatic                                  | 1   | 1   | 1   |
| Oligosymptomatic                              | 1.26 (0.88–1.81)  | 2.06 (1.36–3.12)  | 2.11 (1.39–3.20)  |
| Symptomatic                                   | 2.04 (1.36–3.05)  | 2.05 (1.25–3.37)  | 1.47 (0.89–2.41)  |
|   | $N = 254$ Wald $\chi^2 = 11.8$ , $df = 2$ P = 0.0027                | N = 185 Wald $\chi^2$ = 14.2, df = 2 P = 0.0008                 | N = 173 Wald $\chi^2$ = 12.3, df = 2 P = 0.0021                 |

Coefficients expressed as incidence rate ratios. Data fitted to random effects negative binomial regression models, with time since infection as a covariate. <sup>1</sup>Model includes In bone marrow parasite number as a covariate.

<sup>2</sup>Infectiousness to sandflies measured by xenodiagnosis.

<sup>3</sup>Dogs were classified according to their total clinical score as asymptomatic (scores 0–2), oligosymptomatic (3–6) and symptomatic (>6) at each bimonthly examination. doi:10.1371/journal.pntd.0002583.t002



**Figure 2.** Average *L. infantum* parasite loads in dog tissues with time from infection. Parasite loads shown as  $log_{10}$  values for bone marrow (solid line, circles) and ear skin (solid line, triangles) biopsies. Anti-*Leishmania* IgG antibody units (dotted line, open circles) shown for comparison.

doi:10.1371/journal.pntd.0002583.g002

infection; the strongest predictor of being infectious was ear skin parasite load (Table 3); similar results were seen when analysis was restricted to only paired bone marrow and ear skin samples (data not shown). Infectivity to sandflies was associated with high parasite loads in ear skin (Figure 4): the majority of dogs had loads  $<10^6$  parasites per gram and were very rarely infectious. Highly infectious dogs had higher mean parasite loads than mildly infectious dogs (ears: Wald  $\chi^2 = 7.36$ , P = 0.0073; marrow:

 $\chi^2 = 7.21$ , P=0.0067), the latter showing greater average loads than non-infectious dogs (ears:  $\chi^2 = 13.35$ , P=0.0003; marrows:  $\chi^2 = 14.56$ , P=0.0001) (Figure 5).

# Leishmania loads in crab-eating foxes

L. infantum was detected in bone marrow of 50% (17/34) and skin of 67% (20/30) of infected foxes. Parasite loads showed similar over-dispersion as for dogs (Table 1). Of the total L. infantum loads recorded in bone marrows, 90% was attributed to 8% (5/67) of samples and to 12% (4/34) of the foxes. The equivalent figures for skin biopsies were 8% (4/53) of samples and 10% (3/30) of foxes.

Bone marrow loads varied significantly with fox age, rising rapidly in the first 6 months of life (age (months): IRR = 1.25 (95% CL 1.11– 1.42), P = 0.0004) and declining thereafter (months<sup>2</sup>: IRR = 0.997) (0.995-0.999), P = 0.0015); a similar, though not significant, pattern was seen for ear skin samples (P = 0.25) (Supplementary Figure S1). No parasites were detected in 15 bone marrow samples from 6 foxes over 6 years old, whereas 4/6 of these foxes (4/12 samples) showed residual parasites in ear skin. In contrast, anti-Leishmania IgG titres did not decline in older age classes (Supplementary Figure S1). There were significant positive relationships between fox tissue parasite numbers and anti-Leishmania IgG titres (marrow Wald  $\chi^2 = 16.0$ , df = 1, P = 0.0001; skin Wald  $\chi^2$  = 5.68, df = 1, P = 0.017), and ear skin to bone marrow parasite ratios were moderately higher in foxes with high titres (Wald  $\chi^2 = 3.81$ , df = 1, P = 0.051). Only one fox showed any clinical signs of disease (alopecia) but which was mild and transitory.

#### Comparison of dogs and crab-eating foxes

Skin and bone marrow parasite loads of foxes were similar to those in non-infectious dogs (P>0.10) (Figure 5). Seven long-term "truly" asymptomatic infected dogs were identified: they transmitted infection to 1/678 sandflies exposed in 24 xenodiagnosis trials on 4 dogs. Their parasite loads were similar to those in foxes (P>0.18), which were all asymptomatic by the same definition



**Figure 3.** *L. infantum* **parasite load model fits for individual dogs with time from infection.** Lines represent fitted slopes of log<sub>10</sub> parasite loads in (A) bone marrow (per mL) and (B) ear skin (per gram) examined using a multilevel mixed-effects time series model with random intercepts (initial loads) and slopes (parasite loads with time). doi:10.1371/journal.pntd.0002583.g003

**Table 3.** Relationships between the probability of dogs being infectious to sandflies and immunologic, parasitological or clinical parameters.

| Variable                              | n   | OR    | Ρ      | r²    | AUC   |
|---------------------------------------|-----|-------|--------|-------|-------|
| Time since infection (days)           | 103 | 1.003 | 0.0369 | 0.044 | 0.649 |
| Anti-Leishmania antibody <sup>1</sup> | 103 | 5.685 | 0.0001 | 0.296 | 0.857 |
| Total clinical score                  | 93  | 1.285 | 0.0003 | 0.138 | 0.729 |
| Dermatitis score                      | 93  | 1.492 | 0.0006 | 0.127 | 0.708 |
| Other clinical score                  | 93  | 1.376 | 0.0065 | 0.079 | 0.672 |
| PCR                                   | 103 | 26.1  | 0.002  | 0.248 | 0.712 |
| Bone marrow load <sup>1</sup>         | 103 | 2.224 | 0.0001 | 0.258 | 0.837 |
| Ear skin load <sup>1</sup>            | 58  | 2.78  | 0.0001 | 0.488 | 0.937 |

The probability of being infectious to sandflies measured by xenodiagnosis of infected dogs. OR odds ratio; r<sup>2</sup> logistic regression pseudo; AUC area under receiver operator curve; dermatitis score: score of clinical dermatitis in skin only; PCR: negative or positive by standard PCR;

<sup>1</sup>log<sub>10</sub> transformed anti-*Leishmania* antibody units (per mL) measured by ELISA, and *L. infantum* numbers in bone marrow (per mL), and ear skin biopsies (per gram), measured by gPCR.

doi:10.1371/journal.pntd.0002583.t003

(Figure 5). None of the 22 infected foxes tested were infectious in 39 xenodiagnosis trials. Applying the model coefficients from analysis of dog infectivity (Table 2) to fox ear skin parasite data

(n = 53), foxes were predicted to have been infectious with  $\geq 15\%$  probability ( $\geq 10^{4.64}$  parasites/gram in skin) on 6 of 53 occasions for 4 foxes, equivalent to a total predicted number of infectious samples of 2.9 of 53, compared to the observed 0/39 xenopositive trials of infected foxes.

#### Detecting infectious dogs based on Leishmania loads

The performances of qPCR and ELISA to differentiate dogs of different infectious status in the mixed population were tested using positivity threshold values calculated by ROC analysis of the point xenodiagnosis data (see Methods). PCRbased diagnostic tests showed a high sensitivity (94-100%) to detect highly infectious dogs, though the sensitivities of serology-based tests were somewhat lower (78-100%) (Table 4). The sensitivities of most tests to detect mildly infectious dogs were lower, but these dogs contributed <20%of transmission. Only tests based on qPCR thresholds showed high specificities for infectious dogs (i.e. low sensitivities to detect non-infectious dogs) (Table 4). Highly infectious dogs were detected by qPCR significantly earlier after patent infection (152 days [95% CI: 117-186]) than either mildly infectious dogs (442 days [302-582]) or non-infectious dogs (435 days [317–553]) (log rank tests: qPCR:  $\chi^2 > 17.3$ , P<0.0003); estimates for the latter two groups were statistically indistinguishable (P = 0.70). Detection time of highly infectious dogs approximated their observed time to becoming infectious (134 days [68-201]).



**Figure 4. Relationship between** *L. infantum* **parasite loads in infected dog skin biopsies and infectiousness to sandflies.** Individual dog infectiousness was measured as the proportion of exposed and blood-fed colony-reared *Lu. longipalpis* sandflies infected during xenodiagnosis. The figure shows results for individual samples (n = 58, triangles), and for the mean parasite load and proportion infected for each dog (n = 26, circles). Symbol size corresponds to sampling weight (number of fully engorged sandflies dissected). doi:10.1371/journal.pntd.0002583.g004



log<sub>10</sub> L. infantum parasites

**Figure 5.** Average *L. infantum* parasite loads in dog and fox tissues. Parasite loads in bone marrow (light bars) and ear skin (dark bars) biopsy samples measured by real-time qPCR and expressed as mean (and 95% C.l.s)  $log_{10}$  parasites per mL (bone marrow) or per gram (ear skin). Samples from infected dogs are stratified by infectious status. Clinically asymptomatic dogs and foxes presented total clinical scores <3 at all clinical examinations over  $\geq$ 8 months post infection follow-up. Infectiousness categories: highly infectious (n = 24 bone marrow samples; n = 9 ear skin samples); mildly infectious (n = 44, 26); uninfectious (n = 85, 60); asymptomatic dogs (n = 37, 27). Foxes were all asymptomatic (n = 67, 51). doi:10.1371/journal.pntd.0002583.g005

| Table  | 4. Sensitivity and | l specificity | of diagnostic | tests to |
|--------|--------------------|---------------|---------------|----------|
| detect | dogs of different  | infectious    | status.       |          |

|   | Proportion                        | of complex po                            | sitive (n/total)   |  |  |  |
|---|-----------------------------------|--|--------------------|--|--|--|
|   | Proportion                        | Proportion of samples positive (n/total) |                    |  |  |  |
|   | Highly<br>infectious <sup>1</sup> | Mildly<br>infectious                     | Non-<br>infectious |  |  |  |
| Test thresholds to detect infectivity <sup>2</sup>  |                                   |  |                    |  |  |  |
| Ear cut-off   | 1.00 (8/8)                        | 0.50 (14/28)                             | 0.02 (1/64)        |  |  |  |
| BM cut-off  | 0.94 (17/18)                      | 0.35 (16/46)                             | 0.19 (15/79)       |  |  |  |
| ELISA cut-off                                       | 0.82 (27/33)                      | 0.38 (29/76)                             | 0.32 (44/137)      |  |  |  |
| Conventional tests to detect infection <sup>3</sup> |                                   |  |                    |  |  |  |
| Ear positive  | 1.00 (8/8)                        | 1.00 (28/28)                             | 0.78 (50/64)       |  |  |  |
| BM positive   | 1.00 (18/18)                      | 0.83 (38/46)                             | 0.81 (64/79)       |  |  |  |
| ELISA positive                                      | 0.97 (32/33)                      | 0.83 (63/76)                             | 0.87 (120/138)     |  |  |  |
| PCR positive  | 1.00 (24/24)                      | 0.58 (32/55)                             | 0.48 (40/84)       |  |  |  |
| rK39 positive                                       | 0.79 (22/28)                      | 0.57 (34/60)                             | 0.50 (49/98)       |  |  |  |

<sup>1</sup>Dogs were classified as highly infectious, infectious, and non-infectious to the sandfly vector *Lu. longipalpis* by longitudinal xenodiagnosis follow-up. <sup>2</sup>Test performance based on parasite numbers, or IgG antibody units, calculated

at point xenodiagnosis to detect infectiousness of infected dogs.

<sup>3</sup>Test performance based on conventional cut-offs to detect infection.

doi:10.1371/journal.pntd.0002583.t004

## Discussion

We demonstrate pronounced heterogeneity in L. infantum loads between dogs, assessed by qPCR in bone marrow and ear skin. Loads were highly over-dispersed with evidence of greater aggregation in ear skin relative to bone marrow (9% vs 16% of dogs harboured 90% of total parasites). Parasite loads in the two tissues showed different dynamics: bone marrow loads increased rapidly reaching a peak 100-200 days after infection, while ear skin loads continued to increase over a 600 day period, resulting in increased skin to bone marrow load ratios in late infection. Dissemination to the skin varied between dogs, being greater in sick and infectious dogs. Evidence of L. infantum parasite overdispersion has been reported in different dog tissues [27,40-42] and in human blood [15,26,43], and greater variation in parasite loads in ear skin compared to paired bone marrows, lymph nodes, blood, and liver and spleen samples has been reported for Brazilian dogs [40,42]. However these studies did not evaluate parasite loads through time. One cohort study of Italian dogs noted a decrease in ear to lymph node parasite ratios during clinical development, in apparent contrast to results here. In that study, the time of infection was not established, so dogs may have been at a different stage and severity of infection [34].

Tissue parasite load, particularly in ear skin, was the best predictor of being currently infectious to vectors. *L. infantum* amastigotes in skin tissue or skin capillaries are directly accessible to sandflies, which are known to feed abundantly on ear pinnae; and ear skin appears to be more infective than abdomen skin [23,30]. Some of the variation in parasite loads between ear tissue samples may also reflect small scale spatial variation in parasite density within the ear. We did not restrict sandflies to feed only on ears, unlike other studies [23,29,44]. However, the proportion of dogs that were infectious was substantially lower than the proportion with detectable skin parasites, and only dogs with very high skin parasite loads were consistently infectious. Highly infectious dogs showed greater average loads compared to mildly infectious and non-infectious dogs, and also tended to fall within the top 20% parasite loads for each tissue. These data, and the observed high degree of parasite aggregation in ear skin, suggest that the majority of transmission events to vectors result from a small proportion of infectious dogs. Previously we reported for these dogs that 7 of 42 infectious dogs (17%) were responsible for >80% of all sandfly infections [22]. Similar over-dispersion in infectiousness can be calculated from published xenodiagnosis studies, with 15% to 44% of dogs accounting for >80% of transmission events [23,44]. qPCR studies of canid tissue L. infantum loads relative to xenodiagnoses are not available elsewhere, but parasite estimates by immunohistochemistry of ear skin show moderate correlations with xenodiagnosis positivity [30,45]. Our current results suggest that high parasite loads in dog ear skin, rather than the simple presence of parasites, is the important metric to identify likely infectious individuals and potential reservoir populations. In the current study, all infections were shown to be L. infantum [36]. To identify super-spreaders in regions of mixed Leishmania co-infections, the specificity of qPCR methods would need to be fully validated.

Current ZVL control strategy in Brazil includes mass test-andslaughter of Leishmania antibody positive dogs [17], which is criticised on theoretical, logistical and also on ethical grounds [18–22]. If the small fraction of dogs that are responsible for the majority of transmission could be identified (e.g. by detection of high parasite loads) and targeted, this would directly address many of these issues, and may be more cost-effective than mass interventions [9,12]. Canine infectiousness to sandflies is known to increase with the severity of disease and high anti-parasite antibody, but sensitive and specific markers of infectiousness have not been identified [22,23,29,30,46]. Here, we show that adopting quantitative test threshold values based on skin parasite numbers, highly infectious dogs can be distinguished from noninfectious dogs. These tests were highly sensitive for highly infectious dogs, equivalent to detection of 87-94% of sandfly infections in these samples (data not shown), and importantly also showed high specificities (0.83-0.99) to detect non-infectious dogs, unlike conventional tests for infection. Since up to 50% of seropositive dogs may be asymptomatic in a single community survey, such a targeted approach should also raise dog-owner compliance.

## References

- 1. Soper GA (1939) The curious career of Typhoid Mary. Bull N Y Acad Med 15: 698–712.
- Stein RA (2011) Super-spreaders in infectious diseases. Int J Infect Dis 15: e510–e513.
   Anderson RM, May RM (1979) Population Biology of Infectious Diseases 1.
- Nature 280: 361–367.
  4. May RM, Anderson RM (1979) Population Biology of Infectious Diseases 2. Nature 280: 455–461.
- Galvani AP (2003) Immunity, antigenic heterogeneity, and aggregation of helminth parasites. J Parasitol 89: 232–241.
- Shaw DJ, Dobson AP (1995) Patterns of macroparasite abundance and aggregation in wildlife populations: A quantitative review. Parasitology 111: S111–S133.
- Woolhouse MEJ, Dye C, Etard JF, Smith T, Charlwood JD, et al. (1997) Heterogeneities in the transmission of infectious agents: Implications for the design of control programs. Proc Natl Acad Sci U S A 94: 338–342.

The crab-eating fox occurs widely in South America, and is commonly infected with L. infantum [16,35,47], and thus often assumed to be a sylvatic reservoir. However, few infected foxes have been shown to infect sandflies [48,49], and in our cohort study none of the foxes were infectious [35]. Here, we show that fox parasite loads, though heterogeneous, were significantly lower than those of infectious dogs, and similar to non-infectious dogs, providing further evidence that foxes are not likely to be important for maintaining transmission [22,35]. The results also provide a parasitological explanation for why the foxes here, and probably wild canids more generally, tend to present asymptomatic infections [16,50,51]. Relatively low parasite loads were also noted in the truly asymptomatic cohort dogs, as also reported in asymptomatic human infection [26,28,52]. Whether asymptomatic human infections with L. donovani is associated with low parasite loads and thus low transmission potential remains speculative, and further studies are needed [16]. Variation in parasite load between individuals of other potential reservoir hosts (e.g. hares in Iberia [53]), and variation in parasite load in skin between different parts of the host, would also be informative.

In conclusion, this study highlights the importance of quantifying heterogeneities in infection loads in relation to transmission potential through prospective studies, underpinning development of novel tools for parasitic disease management. Studies are now needed to confirm the efficacy of diagnostic threshold-based driven actions against transmission, and to develop diagnostic kits, based on the detection of parasite DNA (e.g. isothermal amplification) or parasite antigens, for practical field use.

#### **Supporting Information**

Figure S1 Average *L. infantum* parasite loads in fox tissues with increasing fox age. Average  $\log_{10} L$ . *infantum* parasite loads in ear skin biopsies (per gram) (solid line, triangles) and bone marrow aspirates (per mL) (solid line, circles) with fox ageclass in a naturally infected crab-eating fox population. Also shown are  $\log_{10}$  anti-*Leishmania* IgG antibody units (per mL) (dotted line, open circles) for comparison. Data are shown for infected foxes only. (TIF)

#### Acknowledgments

The *L. infantum* strain for real-time qPCR was supplied by Dr. A. Tomas (Institute for Molecular and Cell Biology, Porto, Portugal). We thank Professor G.F. Medley (WIDER) and three anonymous referees for comments on earlier drafts of the manuscript.

## **Author Contributions**

Conceived and designed the experiments: OC CC RJQ. Performed the experiments: OC CC RJQ LMG LCB. Analyzed the data: OC CC RJQ. Contributed reagents/materials/analysis tools: OC CC RJQ LMG LCB. Wrote the paper: OC CC RJQ.

- Matthews L, Low JC, Gally DL, Pearce MC, Mellor DJ, et al. (2006) Heterogeneous shedding of *Escherichia coli* O157 in cattle and its implications for control. Proc Natl Acad Sci U S A 103: 547–552.
- Lloyd-Smith JO, Schreiber SJ, Kopp PE, Getz WM (2005) Superspreading and the effect of individual variation on disease emergence. Nature 438: 355–359.
- Lau LLH, Ip DKM, Nishiura H, Fang VJ, Chan K-H, et al. (2013) Heterogeneity in viral shedding among individuals with medically attended influenza a virus infection. J Infect Dis 207: 1281–1285.
- Eames KTD, Keeling MJ (2002) Modeling dynamic and network heterogeneities in the spread of sexually transmitted diseases. Proc Natl Acad Sci U S A 99: 13330–13335.
- Dowdy DW, Golub JE, Chaisson RE, Saraceni V (2012) Heterogeneity in tuberculosis transmission and the role of geographic hotspots in propagating epidemics. Proc Natl Acad Sci U S A 109: 9557–9562.

- Fraser C, Hollingsworth TD, Chapman R, de Wolf F, Hanage WP (2007) Variation in HIV-1 set-point viral load: Epidemiological analysis and an evolutionary hypothesis. Proc Natl Acad Sci U S A 104: 17441–17446.
- Chase-Topping M, Gally D, Low C, Matthews L, Woolhouse M (2008) Supershedding and the link between human infection and livestock carriage of *Escherichia coli* O157. Nature Rev Microbiol 6: 904–912.
- Mary C, Faraut F, Drogoul M-P, Xeridat B, Schleinitz N, et al. (2006) Reference values for *Leishmania infantum* parasitemia in different clinical presentations: Quantitative polymerase chain reaction for therapeutic monitoring and patient follow-up. Am J Trop Med Hyg 75: 858–863.
- Quinnell RJ, Courtenay O (2009) Transmission, reservoir hosts and control of zoonotic visceral leishmaniasis. Parasitology 136: 1915–1934.
- Ministério da Saúde (2006) Manual de vigilância e controle da leishmaniose visceral. Brasília: Editora do Ministério da Saúde. 120 p.
- Costa CHN (2011) How effective is dog culling in controlling zoonotic visceral leishmaniasis? A critical evaluation of the science, politics and ethics behind this public health policy. Rev Soc Bras Med Trop 44: 232–242.
- Maia-Elkhoury ANS, Alves WA, de Sousa-Gomes ML, de Sena JM, Luna EA (2008) Visceral leishmaniasis in Brazil: trends and challenges. Cad Saúde Pública 24: 2941–2947.
- Oliveira CDL, Franco Morais MH, Lins Machado-Coelho GL (2008) Visceral leishmaniasis in large Brazilian cities: challenges for control. Cad Saúde Pública 24: 2953–2958.
- Esch KJ, Pontes NN, Arruda P, O'Connor A, Morais L, et al. (2012) Preventing zoonotic canine leishmaniasis in northeastern Brazil: pet attachment and adoption of community *Leishmania* prevention. Am J Trop Med Hyg 87: 822–831.
- Courtenay O, Quinnell RJ, Garcez LM, Shaw JJ, Dye C (2002) Infectiousness in a cohort of Brazilian dogs: Why culling fails to control visceral leishmaniasis in areas of high transmission. J Infect Dis 186: 1314–1320.
- Travi BL, Tabares CJ, Cadena H, Ferro C, Osorio Y (2001) Canine visceral leishmaniasis in Colombia: Relationship between clinical and parasitologic status and infectivity for sand flies. Am J Trop Med Hyg 64: 119–124.
- Manna L, Reale S, Vitale F, Gravino AE (2009) Evidence for a relationship between *Leishmania* load and clinical manifestations. Res Vet Sci 87: 76–78.
- Bossolasco S, Gaiera G, Olchini D, Gulletta M, Martello L, et al. (2003) Realtime PCR assay for clinical management of human immunodeficiency virusinfected patients with visceral leishmaniasis. J Clin Microbiol 41: 5080–5084.
- Mary C, Faraut F, Lascombe L, Dumon H (2004) Quantification of *Leishmania* infantum DNA by a real-time PCR assay with high sensitivity. J Clin Microbiol 42: 5249–5255.
- Francino O, Altet L, Sanchez-Robert E, Rodriguez A, Solano-Gallego L, et al. (2006) Advantages of real-time PCR assay for diagnosis and monitoring of canine leishmaniosis. Vet Parasitol 137: 214–221.
- Costa CHN, Gomes RBB, Silva MRB, Garcez LM, Ramos PKS, et al. (2000) Competence of the human host as a reservoir for *Leishmania chagasi*. J Infect Dis 182: 997–1000.
- Michalsky EM, Fonseca Rocha M, Mariano da Rocha Lima ACV, Carlos Franca-Silva J, Quinhone Pires M, et al. (2007) Infectivity of seropositive dogs, showing different clinical forms of leishmaniasis, to *Lutzomyia longipalpis* phlebotomine sand flies. Vet Parasitol 147: 67–76.
- Vercosa BLA, Lemos CM, Mendonca IL, Silva SMMS, de Carvalho SM, et al. (2008) Transmission potential, skin inflammatory response, and parasitism of symptomatic and asymptomatic dogs with visceral leishmaniasis. BMC Vet Res 4: 45.
- Quinnell RJ, Courtenay O, Garcez L, Dye C (1997) The epidemiology of canine leishmaniasis: transmission rates estimated from a cohort study in Amazonian Brazil. Parasitology 115: 143–156.
- 32. Quaresma PF, Fonseca Murta SM, Ferreira EdC, Vianna Mariano da Rocha-Lima AC, Prates Xavier AA, et al. (2009) Molecular diagnosis of canine visceral leishmaniasis: Identification of Leishmania species by PCR-RFLP and quantification of parasite DNA by real-time PCR. Acta Trop 111: 289– 294.
- 33. Giunchetti RC, Mayrink W, Carneiro CM, Correa-Oliveira R, Martins-Filho OA, et al. (2008) Histopathological and immunohistochemical investigations of the hepatic compartment associated with parasitism and serum biochemical changes in canine visceral leishmaniasis. Res Vet Sci 84: 269–277.
- Manna L, Reale S, Viola E, Vitale F, Manzillo VF, et al. (2006) *Leishmania* DNA load and cytokine expression levels in asymptomatic naturally infected dogs. Vet Parasitol 142: 271–280.

- Courtenay O, Quinnell RJ, Garcez LM, Dye C (2002) Low infectiousness of a wildlife host of *Leishmania infantum*: the crab-eating fox is not important for transmission. Parasitology 125: 407–414.
- Quinnell RJ, Courtenay O, Davidson S, Garcez L, Lambson B, et al. (2001) Detection of *Leishmania infantum* by PCR, serology and cellular immune response in a cohort study of Brazilian dogs. Parasitology 122: 253–261.
- Quinnell RJ, Carson C, Reithinger R, Garcez LM, Courtenay O (2013) Evaluation of rK39 rapid diagnostic tests for canine visceral leishmaniasis: longitudinal study and meta-analysis. PLoS Negl Trop Dis 7: e1992.
- Carson C, Quinnell RJ, Holden J, Garcez LM, Deborggraeve S, et al. (2010) Comparison of *Leishmania* OligoC-TesT PCR with conventional and real-time PCR for diagnosis of canine *Leishmania* infection. J Clin Microbiol 48: 3325– 3330.
- Klein JP, Moeschberger ML (1997) Survival Analysis: Techniques for Censored and Truncated data. New York: Springer.
- 40. Ferreira SdA, Leite RS, Ituassu LT, Almeida GG, Souza DM, et al. (2012) Canine skin and conjunctival swab samples for the detection and quantification of *Leishmania infantum* DNA in an endemic urban area in Brazil. PLoS Negl Trop Dis 6: e1596.
- 41. Mohammadiha A, Haghighi A, Mohebali M, Mandian R, Abadi AR, et al. (2013) Canine visceral leishmaniasis: a comparative study of real-time PCR, conventional PCR, and direct agglutination on sera for the detection of *Leishmania infantum* infection. Vet Parasitol 192: 83–90.
- 42. Lima LVR, Carneiro LA, Campos MB, Chagas EJ, Laurenti MD, et al. (2010) Canine visceral leishmaniasis due to *Leishmania* (*L*) *infantum chagasi* in Amazonian Brazil: comparison of the parasite density from the skin, lymph node and visceral tissues between symptomatic and asymptomatic, seropositive dogs. Rev Inst Med Trop Sao Paulo 52: 259–265.
- 43. Mohammadiha A, Mohebali M, Haghighi A, Mandian R, Abadi AR, et al. (2013) Comparison of real-time PCR and conventional PCR with two DNA targets for detection of *Leishmania (Leishmania) infantum* infection in human and dog blood samples. Exp Parasitol 133: 89–94.
- Molina R, Amela C, Nieto J, Sanandres M, Gonzalez F, et al. (1994) Infectivity of dogs naturally infected with *Leishmania infantum* to colonised *Phlebotomus* periiciosus. T Roy Soc Trop Med H 88: 491–493.
- 45. Amorim IFGd, Silva SMd, Figueiredo MM, Moura EP, Castro RSd, et al. (2011) Toll receptors type-2 and CR3 expression of canine monocytes and its correlation with immunohistochemistry and xenodiagnosis in visceral leishmaniasis. PLoS One 6: e27679.
- 46. da Costa-Val AP, Cavalcanti RR, Gontijo NdF, Marques Michalick MS, Alexander B, et al. (2007) Canine visceral leishmaniasis: Relationships between clinical status, humoral immune response, haematology and *Lutzomyia (Lutzomyia) longipalpis* infectivity. Vet J 174: 636–643.
- Courtenay O, Macdonald DW, Lainson R, Shaw JJ, Dye C (1994) Epidemiology of canine leishmaniasis- a comparative serological study of dogs and foxes in Amazon Brazil. Parasitology 109: 273–279.
- 48. Lainson R, Dye C, Shaw JJ, Macdonald DW, Courtenay O, et al. (1990) Amazonian visceral leishmaniasis- distribution of the vector *Lutzomyia longipalpis* (Lutz and Neiva) in relation to the fox *Cerdocyon thous* (Linn) and the efficiency of this reservoir host as a source of infection. Mem Inst Oswaldo Cruz 85: 135–137.
- Gomes RB, Mendonca IL, Silva VC, Ruas J, Silva MB, et al. (2007) Antibodies against *Lutzomyia longipalpis* saliva in the fox *Cerdocyon thous* and the sylvatic cycle of *Leishmania chagasi*. T Roy Soc Trop Med H 101: 127–133.
- Verin R, Poli A, Ariti G, Nardoni S, Fanucchi MB, et al. (2010) Detection of Leishmania infantum DNA in tissues of free-ranging red foxes (Vulpes vulpes) in central Italy. Eur J Wildlife Res 56: 689–692.
- Dipineto L, Manna L, Baiano A, Gala M, Fioretti A, et al. (2007) Presence of Leishmania infantum in red foxes (Vulpes vulpes) in southern Italy. J Wildlife Dis 43: 518–520.
- 52. Marques LHdS, Gomes LI, Marques da Rocha IC, Marques da Silva TA, Oliveira E, et al. (2012) Low parasite load estimated by qPCR in a cohort of children living in urban area endemic for visceral leishmaniasis in Brazil. PLoS Negl Trop Dis 6: e1955.
- Molina R, Jimenez MI, Cruz I, Iriso A, Martin-Martin I, et al. (2012) The hare (*Lepus granatensis*) as potential sylvatic reservoir of *Leishmania infantum* in Spain. Vet Parasitol 190: 268–271.